\begin{table}%t3
\caption{Observations of the sources in the CORALZ sample. The first 18 objects form the CORALZ core sample at $0.005<z<0.160$, the next seven objects are other nearby radio sources in the CORALZ sample, and the final three objects are other compact radio sources located towards probably random foreground galaxies. \label{TCOR_1}}
\small%\centering
\par
\begin{tabular}{ccccccrccr}
\hline\hline
IAU Name & Fig. & Epoch & Freq. & Bandw. & Array & $S_{\rm{peak}}$                       &
 rms                       & \multicolumn{2}{c}{Beam} \\
         &      &       & (GHz) & (MHz)  &       & \multicolumn{2}{c}{(mJy beam$^{-1}$)} &
 \multicolumn{1}{c}{(mas)} & $\degr$                  \\
\hline
\multicolumn{8}{l}{The CORALZ core sample} \\
\object{J073328+560541} &~\ref{FCOR_1} & 2000/02/14 & 1.655 & 32 & EVN      & 126 & 0.39 &  25 $\pm$  15 & 31 \\
                        &~\ref{FCOR_1} & 2004/06/05 & 1.665 & 16 & EVN+VLBA &  55 & 0.22 & 7.8 $\pm$ 2.1 & --12 \\
                        &~\ref{FCOR_1} & 2004/05/24 & 4.993 & 16 & EVN+VLBA &  47 & 0.19 & 9.0 $\pm$ 7.9 & --59 \\
\object{J073934+495438} &~\ref{FCOR_2} & 2000/03/02 & 4.987 & 32 & EVN+VLBA &  22 & 0.06 & 2.8 $\pm$ 0.8 & --25 \\
                        &~\ref{FCOR_2} & 2004/05/24 & 4.993 & 16 & EVN+VLBA &  25 & 0.06 & 4.1 $\pm$ 1.3 & --32 \\
\object{J083139+460800} &~\ref{FCOR_2} & 2000/03/02 & 4.987 & 32 & EVN+VLBA &  32 & 0.13 & 2.3 $\pm$ 0.7 & --6 \\
                        &~\ref{FCOR_2} & 2004/05/24 & 4.993 & 16 & EVN+VLBA &  42 & 0.08 & 3.1 $\pm$ 0.8 & --13 \\
\object{J083637+440109} &~\ref{FCOR_3} & 2000/11/13 & 1.658 & 15 & MERLIN   &  22 & 0.09 & 125 $\pm$ 121 & --75 \\
\object{J090615+463618} &~\ref{FCOR_1} & 2001/05/27 & 1.659 & 16 & EVN      & 134 & 0.13 &  22 $\pm$  13 & 59 \\
                        &~\ref{FCOR_1} & 2004/06/05 & 1.665 & 16 & EVN+VLBA &  88 & 0.16 & 9.9 $\pm$ 2.1 & --4 \\
                        &~\ref{FCOR_1} & 2004/05/24 & 4.993 & 16 & EVN+VLBA &  84 & 0.18 & 2.8 $\pm$ 0.7 & --14 \\
\object{J102618+454229} &~\ref{FCOR_1} & 2001/05/28 & 1.659 & 16 & EVN      &  76 & 0.14 &  22 $\pm$  15 & 69 \\
                        &~\ref{FCOR_1} & 2004/06/05 & 1.665 & 16 & EVN+VLBA &  30 & 0.10 &  10 $\pm$ 2.1 & --3 \\
                        &~\ref{FCOR_1} & 2004/05/24 & 4.993 & 16 & EVN+VLBA &  30 & 0.06 & 2.7 $\pm$ 0.8 & --3 \\
\object{J103719+433515} &~\ref{FCOR_1} & 2001/05/27 & 1.659 & 16 & EVN      &  61 & 0.23 &  21 $\pm$  13 & 77 \\
                        &~\ref{FCOR_1} & 2004/06/05 & 1.665 & 16 & EVN+VLBA &  21 & 0.10 &  12 $\pm$ 2.1 & --7 \\
                        &~\ref{FCOR_1} & 2004/05/24 & 4.993 & 16 & EVN+VLBA &  19 & 0.23 &  11 $\pm$ 6.7 & 34 \\
\object{J115000+552821} &~\ref{FCOR_3} & 2001/05/28 & 1.659 & 16 & EVN      & 103 & 0.14 &  20 $\pm$  14 & 63 \\
\object{J120902+411559} &~\ref{FCOR_1} & 2001/05/27 & 1.659 & 16 & EVN      & 117 & 0.17 &  23 $\pm$  14 & 48 \\
                        &~\ref{FCOR_1} & 2004/06/05 & 1.665 & 16 & EVN+VLBA &  54 & 0.05 & 9.5 $\pm$ 2.1 & --11 \\
                        &~\ref{FCOR_1} & 2004/05/24 & 4.993 & 16 & EVN+VLBA &  73 & 0.54 & 9.1 $\pm$ 6.1 & 43 \\
\object{J131739+411545} &~\ref{FCOR_2} & 2000/03/02 & 4.987 & 32 & EVN+VLBA &  45 & 0.22 & 1.4 $\pm$ 0.9 & --25 \\
                        &~\ref{FCOR_2} & 2004/05/24 & 4.993 & 16 & EVN+VLBA &  53 & 0.11 & 2.7 $\pm$ 0.7 & --3 \\
\object{J140051+521606} &~\ref{FCOR_3} & 2000/12/26 & 1.408 & 15 & MERLIN   & 101 & 0.20 & 163 $\pm$ 150 & --82 \\
\object{J140942+360416} &~\ref{FCOR_1} & 2001/05/28 & 1.659 & 16 & EVN      &  81 & 0.13 &  28 $\pm$  11 & 37 \\
                        &~\ref{FCOR_1} & 2004/06/05 & 1.665 & 16 & EVN+VLBA &  30 & 0.06 &  17 $\pm$ 2.1 & --5 \\
\object{J143521+505122} &~\ref{FCOR_3} & 2000/12/26 & 1.408 & 15 & MERLIN   &  73 & 0.26 & 166 $\pm$ 146 & --41 \\
\object{J150805+342323} &~\ref{FCOR_3} & 2002/02/19 & 1.650 & 16 & EVN      &   4 & 0.11 &  24 $\pm$  23 & 16 \\
\object{J160246+524358} &~\ref{FCOR_1} & 2001/05/27 & 1.659 & 16 & EVN      &  62 & 0.14 &  21 $\pm$  12 & 59 \\
                        &~\ref{FCOR_1} & 2004/06/05 & 1.665 & 16 & EVN+VLBA &  23 & 0.16 & 8.4 $\pm$ 2.1 & --19 \\
                        &~\ref{FCOR_1} & 2004/05/24 & 4.993 & 16 & EVN+VLBA &  30 & 0.13 & 7.6 $\pm$ 6.3 & 58 \\
\object{J161148+404020} &~\ref{FCOR_3} & 2000/11/25 & 1.658 & 15 & MERLIN   &  86 & 0.12 & 167 $\pm$ 136 & --1 \\
\object{J170330+454047} \\
\object{J171854+544148} &~\ref{FCOR_1} & 2000/02/14 & 1.655 & 32 & EVN      & 119 & 0.33 &  20 $\pm$  13 & --84 \\
                        &~\ref{FCOR_1} & 2004/06/05 & 1.665 & 16 & EVN+VLBA &  90 & 0.19 &  10 $\pm$ 2.1 & --6 \\
                        &~\ref{FCOR_1} & 2004/05/24 & 4.993 & 16 & EVN+VLBA &  49 & 0.11 & 8.9 $\pm$ 6.1 & 16 \\
\multicolumn{8}{l}{Other nearby sources in the CORALZ sample} \\
\object{J093609+331308} &~\ref{FCOR_4} & 2000/03/02 & 4.987 & 32 & EVN+VLBA &  28 & 0.15 & 1.2 $\pm$ 0.8 & 0 \\
\object{J101636+563926} &~\ref{FCOR_4} & 2000/11/09 & 1.658 & 15 & MERLIN   &  52 & 0.08 & 174 $\pm$ 136 & --31 \\
\object{J105731+405646} &~\ref{FCOR_4} & 2000/03/02 & 4.987 & 32 & EVN+VLBA &  14 & 0.08 & 1.8 $\pm$ 0.3 & 13 \\
\object{J115727+431806} &~\ref{FCOR_4} & 2001/01/02 & 1.408 & 15 & MERLIN   &  41 & 0.09 & 182 $\pm$ 173 & --47 \\
\object{J132513+395552} &~\ref{FCOR_4} & 2000/03/02 & 4.987 & 32 & EVN+VLBA &  15 & 0.08 & 0.7 $\pm$ 0.3 & --20 \\
\object{J134035+444817} &~\ref{FCOR_4} & 2000/03/02 & 4.987 & 32 & EVN+VLBA &  76 & 0.13 & 1.4 $\pm$ 0.9 & --7 \\
\object{J155927+533054} &~\ref{FCOR_4} & 2000/12/18 & 1.408 & 15 & MERLIN   &  16 & 0.11 & 308 $\pm$ 202 & 14 \\
\hline
\multicolumn{8}{l}{Additional sources} \\
\object{J071509+452555} &~\ref{FCOR_5} & 2000/03/02 & 4.987 & 32 & EVN+VLBA &  30 & 0.13 & 0.8 $\pm$ 0.8 &  0 \\
\object{J080454+433537} &~\ref{FCOR_5} & 2000/02/14 & 1.655 & 32 & EVN      & 182 & 0.39 &  27 $\pm$  14 & 39 \\
\object{J134158+541524} &~\ref{FCOR_5} & 2002/02/19 & 1.650 & 16 & EVN      &  10 & 0.13 &  25 $\pm$  22 & 70 \\
\hline
\end{tabular}
\end{table}